Author's response to reviews

Title: Cardiovascular Risk Factors and 30-Year Cardiovascular Risk in Homeless Adults with Mental Illness

Authors:

Agnes Gozdzik (agnes.gozdzik@gmail.com)
Roxana Salehi (RSalehi@uwgt.org)
Patricia O'Campo (O'CampoP@smh.ca)
Vicky Stergiopoulos (Stergiopoulosv@smh.ca)
Stephen W Hwang (HWANGS@smh.ca)

Version: 2
Date: 29 September 2014

Author's response to reviews:

Dear Editorial Staff,

Please see the uploaded revised manuscript which has the following requested changes:

1) Line numbering has been added to the document to facilitate review.
2) Email addresses of all the authors are now included on the manuscript title page.

Sincerely,
Agnes

Gozdzik Ag, PhD | Research Associate
Centre for Research on Inner City Health | Li Ka Shing Knowledge Institute
St. Michael's Hospital | 209 Victoria Street | Third Floor | Toronto, ON M5C 1N8
GozdzikAg@smh.ca | (416) 864-6060 x 77460